Author's response to reviews

Title: An unusual presentation of Castleman's Disease: a case report

Authors:

Dr Ifeoma S Izuchukwu (dshaikhgroupe@aol.com)
Kamal Tourbaf (ktourbaf@ams.ecmc.edu)
Dr Martin C Mahoney (martin.Mahoney@roswellpark.org)

Version: 4 Date: 2 Sep 2003

Dear Editor,
Thank you for all your help so far. The following changes were made since the last submission:
1. Names of authors have no title attached to them.
2. Country of affiliation indicated on the first page
3. Abstract have been structured into Background, Case presentation, conclusions and these subheadings are not underlined.
4. Manuscript is arranged as abstract, background, case presentation, conclusions, competing interests, Author's contributions, Acknowledgements, References with no et al after authors' names, Figure legends with no figures attached to manuscript, tables.

Thank you again for your help. Let me know of any updates after this submission.

Sincerely,
Ifeoma Izuchukwu.